\newcommand{\etalchar}[1]{$^{#1}$}
\begin{thebibliography}{MAB{\etalchar{+}}94}

\bibitem[AD94]{AD94}
R.~Alur and D.L. Dill.
\newblock A theory of timed automata.
\newblock {\em Theor. Comp. Sci.}, 126:183--235, 1994.

\bibitem[AH89]{TPTL1}
R.~Alur and T.A. Henzinger.
\newblock A really temporal logic.
\newblock In {\em Proc. 30th IEEE Symp. Found. of Comp. Sci.}, pages 164--169,
  1989.

\bibitem[AH92]{AHsur}
R.~Alur and T.~Henzinger.
\newblock Logics and models of real time: A survey.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 74--106.
  Springer-Verlag, 1992.

\bibitem[AL94]{AL91}
M.~Abadi and L.~Lamport.
\newblock An old-fashioned recipe for real time.
\newblock {\em ACM Trans. Prog. Lang. Sys.}, 16(5):1543--1571, 1994.

\bibitem[BAMP83]{BMP}
M.~Ben-Ari, Z.~Manna, and A.~Pnueli.
\newblock The temporal logic of branching time.
\newblock {\em Acta Informatica}, 20:207--226, 1983.

\bibitem[BH81]{BH81}
A.~Bernstein and P.~K. Harter.
\newblock Proving real time properties of programs with temporal logic.
\newblock In {\em Proceedings of the Eighth Symposium on Operating Systems
  Principles}, pages 1--11. ACM, 1981.

\bibitem[EC82]{EC}
E.A. Emerson and E.M. Clarke.
\newblock Using branching time temporal logic to synthesize synchronization
  skeletons.
\newblock {\em Sci. Comp. Prog.}, 2:241--266, 1982.

\bibitem[FG95]{FG95}
L.~Fix and O.~Grumberg.
\newblock Verificaiton of temporal properties.
\newblock to appear in {\sl Logic and Computation}, 1995.

\bibitem[Har87]{H}
D.~Harel.
\newblock Statecharts: {A} visual formalism for complex systems.
\newblock {\em Sci. Comp. Prog.}, 8:231--274, 1987.

\bibitem[Hen92]{IPL}
T.A. Henzinger.
\newblock Sooner is safer than later.
\newblock {\em Info. Proc. Lett.}, 43(3):135--142, 1992.

\bibitem[HMP94]{TPMTTS}
T.~Henzinger, Z.~Manna, and A.~Pnueli.
\newblock Temporal proof methodologies for timed transition systems.
\newblock {\em Inf. and Comp.}, 112(2):273--337, 1994.

\bibitem[KdR85]{KdR}
R.~Koymans and W.-P. de~Roever.
\newblock Examples of a real-time temporal logic specifications.
\newblock In B.D. Denvir, W.T. Harwood, M.I. Jackson, and M.J. Wray, editors,
  {\em The Analysis of Concurrent Systems}, volume 207 of {\em Lect. Notes in
  Comp. Sci.}, pages 231--252. Springer-Verlag, 1985.

\bibitem[Koy90]{Koy90}
R.~Koymans.
\newblock Specifying real-time properties with metric temporal logic.
\newblock {\em Real-time Systems}, 2(4):255--299, 1990.

\bibitem[KVdR83]{KVdR}
R.~Koymans, J.~Vytopyl, and W.-P. de~Roever.
\newblock Real-time programming and asynchronous message passing.
\newblock In {\em Proc. 2nd ACM Symp. Princ. of Dist. Comp.}, pages 187--197,
  1983.

\bibitem[Lam95]{possibility}
L.~Lamport.
\newblock Proving possibiity properties.
\newblock Technical Report 137, Digital Equipment Corporation, Systems Research
  Center, Palo Alto, July 1995.

\bibitem[MAB{\etalchar{+}}94]{STeP}
Z.~Manna, A.~Anuchitanukul, N.~Bj{\o}rner, A.~Browne, E.~Chang, M.~Col\'{o}n,
  L.~De Alfaro, H.~Devarajan, H.~Sipma, and T.~Uribe.
\newblock {STeP}: The {S}tanford {T}emporal {P}rover.
\newblock Technical Report STAN-CS-TR-94-1518, Dept. of Comp. Sci., Stanford
  University, Stanford, California, 1994.

\bibitem[MMP92]{MMP}
O.~Maler, Z.~Manna, and A.~Pnueli.
\newblock From timed to hybrid systems.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 447--484.
  Springer-Verlag, 1992.

\bibitem[MP91a]{Comp91}
Z.~Manna and A.~Pnueli.
\newblock Completing the temporal picture.
\newblock {\em Theor. Comp. Sci.}, 83(1):97--130, 1991.

\bibitem[MP91b]{MPB}
Z.~Manna and A.~Pnueli.
\newblock {\em The Temporal Logic of Reactive and Concurrent Systems:
  Specification}.
\newblock Springer-Verlag, New York, 1991.

\bibitem[MP93]{ModR}
Z.~Manna and A.~Pnueli.
\newblock Models for reactivity.
\newblock {\em Acta Informatica}, 30:609--678, 1993.

\bibitem[MP94]{tacs94}
Z.~Manna and A.~Pnueli.
\newblock Temporal verification diagrams.
\newblock In T.~Ito and A.~R. Meyer, editors, {\em Theoretical Aspects of
  Computer Software}, volume 789 of {\em Lect. Notes in Comp. Sci.}, pages
  726--765. Springer-Verlag, 1994.

\bibitem[MP95]{MPB2}
Z.~Manna and A.~Pnueli.
\newblock {\em Temporal Verification of Reactive Systems: Safety}.
\newblock Springer-Verlag, New York, 1995.

\bibitem[MT90]{MT90}
F.~Moller and C.~Tofts.
\newblock A temporal calculus of communicating systems.
\newblock In J.C.M. Baeten and J.W. Klop, editors, {\em Proceedings of
  Concur'90}, volume 458 of {\em Lect. Notes in Comp. Sci.}, pages 401--415.
  Springer-Verlag, 1990.

\bibitem[NSY92]{NSY91}
X.~Nicollin, J.~Sifakis, and S.~Yovine.
\newblock From {ATP} to timed graphs and hybrid systems.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 549--572.
  Springer-Verlag, 1992.

\bibitem[Ost90]{BOST}
J.S. Ostroff.
\newblock {\em Temporal Logic of Real-Time Systems}.
\newblock Advanced Software Development Series. Research Studies Press (John
  Wiley \& Sons), Taunton, England, 1990.

\bibitem[Pnu92]{vital}
A.~Pnueli.
\newblock How vital is liveness?
\newblock In W.R. Cleaveland, editor, {\em Proceedings of Concur'92}, volume
  630 of {\em Lect. Notes in Comp. Sci.}, pages 162--175. Springer-Verlag,
  1992.

\bibitem[SBM92]{SBM91}
F.~B. Schneider, B.~Bloom, and K.~Marzullo.
\newblock Putting time into proof outlines.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 618--639.
  Springer-Verlag, 1992.

\bibitem[Sif91]{Si91}
J.~Sifakis.
\newblock An overview and synthesis on timed process algebra.
\newblock In K.G. Larsen and A.~Skou, editors, {\em 3rd Computer Aided
  Verification Workshop}, volume 575 of {\em Lect. Notes in Comp. Sci.}, pages
  376--398. Springer-Verlag, 1991.

\end{thebibliography}

\section{Verifying Safety Properties of Clocked Transition Systems}
\label{safety}
In this section, we present methods for verifying safety properties of
clocked transitions systems.

\subsection{The Invariance Rule}
A temporal formula $\varphi$ that is valid over a program $P$ specifies
a property of $P$, i.e., states a condition that is satisfied by all
computations of $P$. As is explained in \cite{MPB}, the properties
expressible by temporal logic can be arranged in a hierarchy that
identifies different classes of properties according to the form of
formulas expressing them.

Here we will consider only \emm{invariance} properties, namely,
properties that can be expressed by the formula $\G p$,
for some assertion $p$.

\ssb{The Accessibility Rule}
As a preliminary step, we introduce a rule that establishes the
$\Phi$-validity of an assertion $p$. This is rule \acc, presented in
\Figr{acc}.
\begin{figure}
\LLbox{\boxline{\qquad For assertions $\varp$ and $p$,}%\lskip
\Ruleb{A1.\quad $\varp\ \rar\ p$\cr
\noalign{\vskip 5pt}
A2.\quad $\The\ \rar\ \varp$\cr
\noalign{\vskip 5pt}
A3.\quad $\rho_\tau\band\varp\ \ \rar\ \ \varp\pri$\quad for every
$\tau\in\Taut$\cr
\Bar
$\qquad\hskip 1pt\Phi\vmodels p$\cr}\lskip}
\ctrlined{acc}{Rule \acc\ ($\Phi$-state validity of assertion $p$).}

The rule uses an auxiliary assertion $\varp$.  Premise A1 of rule
\acc\ requires that the auxiliary assertion $\varp$ implies assertion
$p$, whose $\Phi$-state validity we wish to prove.  Premise A2 of the
rule requires that $\The$, the initial condition of $P$, implies the
auxiliary assertion $\varp$.  Premise A3 requires that all transitions
in $\Taut^{\Phi}$ (extended transitions of $\Phi$) preserve
$\varp$. Premises A2 and A3 state that $\varp$ holds at the initial
state of every run and that it propagates from any state to its
$\tau$-successor, for every transition $\tau\Taut$ of the
system. Thus, every state in each run of $\Phi$ satisfies $\varp$. Due
to the implication A1, every such state also satisfies $p$. It follows
that $p$ holds on every accessible state of system $\Phi$ and,
therefore, assertion $p$ is $\Phi$-state valid.

\ssb{Rule \inv}
Assume that we have shown that assertion $p$ is $\Phi$-state valid,
i.e., every accessible state of $\Phi$ satisfies $p$. Since all states
appearing in a computation are accessible (every computation is
a run), all states in every computation satisfy $p$. It follows
that every computation of $\Phi$ satisfies $\G p$. Thus, we may use
the premises of rule \acc\ to establish that the temporal formula $\G
p$ is $\Phi$-valid. Consequently, we propose rule \inv, presented in
\Figr{inv}, as the main tool for verifying invariance properties of a
\ctss\ $\Phi$.
\begin{figure}
\LLbox{\boxline{\qquad For assertions $\varp$ and $p$,}%\lskip
\Ruleb{I1.\quad $\varp\ \rar\ p$\cr
\noalign{\vskip 5pt}
I2.\quad $\The\ \rar\ \varp$\cr
\noalign{\vskip 5pt}
I3.\quad $\rho_\tau\band\varp\ \ \rar\ \ \varp\pri$\quad for every
$\tau\in\Taut$\cr
\Bar
$\qquad\hskip 1pt\Phi\models\boc p$\cr}\lskip}
\ctrlined{inv}{Rule \inv\ (invariance) applied to \ctss\ $\Phi$.}

\begin{example}\quad
We use rule \inv\ to establish the invariance of the
assertion
\[p\colon\quad \atl{0,1}\lvee \atm{1}\]
over program \anyy\ (Fig.~\ref{anyy}). This assertion claims that, at
every state in the execution of program \anyy, either control of
process $P_1$ is at $\ell_0$ or $\ell_1$, or control of
$P_2$ is at $m_1$. In particular, it implies that if $P_1$ is at
$\ell_2$ then $P_2$ has already arrived in $m_1$.

We apply rule \inv\ to this choice of $p$, taking
\[\varp\colon\quad (x=0\wedge \atl{0,1})\lvee \atm{1}\]
as the auxiliary assertion.

Premise I1 assumes the form
\[\underbrace{(x=0 \wedge \atl{0,1}) \vee \atm{1}}_{\varp}
\quad\implies\quad \underbrace{\atl{0,1}\lvee \atm{1}}_p,\]
which is obviously valid.

Premise I2 assumes the form
\[\underbrace{\pi= \{\ell_0, m_0\} \lwedge x=0\lwedge
\cdots}_{\Theta}\quad\to\quad
\underbrace{(x=0 \wedge \atl{0,1}) \vee \cdots}_{\varp} \]
which is obviously valid.

Premise I3 has to be checked for each $\tau\in\{\ell_0, \ell_1,
m_0, \tick\}$. For example, premise I3 for $\ell_0$ assumes the form
$$\begin{array}{l}
\left(\begin{array}{l}
\underbrace{\cdots \wedge
\left(\begin{array}{c}
	x=0\ \wedge\ \pi\pri=\pi-\{\ell_0\}\cup\{\ell_1\} \\
		\vee\\
	x\neq 0\ \wedge\ \pi\pri=\pi-\{\ell_0\}\cup\{\ell_2\}
\end{array}\right)
\lwedge x\pri=x \lwedge \cdots}_{\rho_{\ell_{0}}}\\
\quad\\[-7pt]
\wedge\ \ \underbrace{\bigl(x=0\wedge \elabinp\bigr)\lvee
m_1\in\pi}_{\varp}
\end{array}\right)
\quad\to\quad \hspace*{1.7cm}\\
\quad\\[-7pt]
\multicolumn{1}{r}{
\underbrace{\bigl(x\pri=0\wedge \elabinpp\bigr)\lvee
m_1\in\pi\pri}_{\varp\pri}.}
\end{array}$$
It is not difficult to see that this implication is valid.

This establishes the $P$-validity of the invariant $\G p$, i.e.,
\[ P\models \G(\atl{0,1}\lvee \atm{1}).\corn\]
\end{example}

\nin
For assertions $p$ and $q$ and transition $\tau$, we introduce the
abbreviation
\[\quad\{p\}\tau\{q\}\qquad \mbox{standing for} \qquad \rot \land p\ \to\
q\pri.\]

\ssb{Invariance Verification Diagrams}
An effective way of presenting the verification of an invariance
property by rule \inv\ is provided by the graphical formalism of
\emm{verification diagrams} \cite{tacs94}. An (invariance)
verification diagram is a directed labeled graph, constructed as
follows:
\begin{itemize}
\item \emm{Nodes} in the graph are labeled by assertions $\varphi_1,
\ldots, \varphi_m$. We will often refer to a node by the assertion
labeling it.
\item \emm{Edges} in the graph represent transitions between
assertions. Each edge departs from one node, connects to another, and
is labeled by the name of a transition in the program. We refer to an edge
labeled by $\tau$ as a \emm{$\tau$-edge}.
\item Some of the nodes are 
designated as \emm{initial nodes}. They are annotated by an entry arrow
\mbox{%\input diag8.pic
\setlength{\unitlength}{0.007in}
\begin{picture}(25,%40
25)(0,%5
5)
\put(5,20){\circle*{10}}
\spline(10,20)
(25,20)(25,0)
\path(23,8)(25,0)(27,8)
\end{picture}
}.
\end{itemize}

\noindent For example, in Fig.~\ref{fig:d1} we present a
verification diagram for program \anyy.
\begin{figure}[hbt]
\vspace{-0.5cm}
\begin{center}
%\leavevmode 
\mbox{\input {d1.pstex_t}}
\caption{\label{fig:d1} Verification Diagram $D_1$}
\end{center} 
\vspace{-1cm}
\end{figure} 

\ssb{Verification Conditions for Diagrams}

With each verification diagram, we associate the following \emm{verification
conditions}:

\begin{itemize}
\item Let $\varphi_1$ be the (single) initial node in the diagram. The
verification condition corresponding to premise I2 of rule \inv\ is
given by
\[\Theta\ \to\ \varphi_1.\]
\item Let $\varphi$ be a node in the graph, $\tau$ be a
transition in the program, and let $\varphi_1,\ldots,\varphi_k$ be the
nodes reached by $\tau$-edges departing from $\varphi$.  The case that
no $\tau$-edges depart from $\varphi$, i.e., $k=0$, is also included.
The verification condition associated with $\varphi$ and $\tau$
(corresponding to premise I3) is given by
\[\triple(\varphi,\tau,\varphi \vee \varphi_1 \vee \ldots \vee \varphi_k)\]
\end{itemize}

\noindent For example, the verification conditions associated with
diagram $D_1$ (Fig.~\ref{fig:d1}), are:

$$\begin{array}{lcl}
\mbox{I2:} & \quad &\Theta \to\ x=0 \wedge \atl{0,1}\\[4pt]
\mbox{I3 for $\varp_1$ and $\ell_0:$} & &
\triple(x=0 \wedge \atl{0,1},\ell_0,x=0\wedge \atl{0,1})\\
\mbox{I3 for $\varp_1$ and $\ell_1:$} & &
\triple(x=0 \wedge \atl{0,1},\ell_1,x=0\wedge \atl{0,1})\\
\mbox{I3 for $\varp_1$ and $m_0:$} & &
\triple(x=0 \wedge \atl{0,1},m_0,{(x=0\wedge \atl{0,1}) \lvee \atm{1}})\\
\mbox{I3 for $\varp_1$ and $\tick:$} & &
\triple(x=0 \wedge \atl{0,1},\tick,x=0\wedge \atl{0,1})\\[4pt]
\mbox{I3 for $\varp_2$ and $\ell_0:$} & &
\triple(\atm{1},\ell_0,\atm{1})\\
\mbox{I3 for $\varp_2$ and $\ell_1:$} & &
\triple(\atm{1},\ell_1,\atm{1})\\
\mbox{I3 for $\varp_2$ and $m_0:$} & &
\triple(\atm{1},m_0,\atm{1})\\
\mbox{I3 for $\varp_2$ and $\tick:$} & &
\triple(\atm{1},\tick,\atm{1})\\
\end{array}$$

We say that a verification diagram is \emm{valid} if all the
verification conditions associated with the diagram are valid.

Let $D_{_P}$ be a verification diagram associated with a
program $P$. Let
$\varphi_1,\ldots,\varphi_m$ be the assertions labeling the nodes of
$D_{_P}$.

\begin{claim}
If the verification diagram $D_P$ is valid, then
\[P \models \G\bigvee_{i = 1}^{m} \varphi_i \]
If, in addition, $\varphi_i \to\ p$ for every $i = 1,\ldots, m$, then
\[P \models \G p.\]
\end{claim}

For example, all the verification conditions associated with
diagram $D_1$ are valid and the verification diagram $D_1$
establishes
\begin{equation}
\label{anyyeq}
\mbox{\sc any-y} \models (\atl{0,1} \lvee \atm{1})
\end{equation}

\subsubsection*{Encapsulation Conventions}
There are several encapsulation conventions that improve the
presentation and readability of verification diagrams. We extend the
notion of a directed graph into a structured directed graph by
allowing \emm{compound nodes} that may encapsulate other nodes, and
edges that may depart or arrive at compound nodes. A node that does
not encapsulate other nodes is called a \emm{basic node}.


We use the following conventions:
\begin{itemize}
\item \emm{Labels of compound nodes}:\quad A diagram containing a
compound node $n$, labeled by an assertion $\GF$ and encapsulating
nodes $n_1,\ldots,n_k$ with assertions $\GF_1,\ldots,\GF_k$, is
equivalent to a diagram in which $n$ is unlabeled and nodes
$n_1,\ldots,n_k$ are labeled by $\GF_1\wedge \GF$, \ldots, and
$\GF_k\wedge \GF$.
\item \emm{Edges entering and exiting compound nodes}:\quad A diagram
containing an edge $e$ connecting node $A$ to a compound node $n$
encapsulating nodes $n_1,\ldots,n_k$ is equivalent to a diagram in
which there is an edge connecting $A$ to each $n_i$, $i=1,\ldots,k$,
with the same label as $e$. Similarly, an edge $e$ connecting
the compound node $n$ to node $B$ is the same as having a separate
edge connecting each $n_i$, $i=1,\ldots,k$, to $B$ with the same label
as $e$.
\end{itemize}

\noindent  With these conventions we can draw a more detailed verification
diagram 
establishing~(\ref{anyyeq}), as shown in Fig.~\ref{fig:d2}.

\begin{figure}[hbt]
\begin{center}
\leavevmode 
\input d2.pstex_t
\caption{\label{fig:d2} A more detailed diagram, using encapsulation
conventions}
\end{center} 
\end{figure}
In the diagram of Fig.~\ref{fig:d2}, the single assertion $x=0\wedge
\atl{0,1}$, has been broken into the two sub-cases: $x=0\wedge \atl{0}$
and $x=0\wedge \atl{1}$, explicitly displaying the fact that
transitions $\ell_0$ and $\ell_1$ cause the system to move between
these two sub-cases.


%% Old %%
We also use invariance diagrams to provide succinct representations of
proofs by rule \inv.

\begin{example}\quad
We use rule \inv\ to prove that program \anyce\ terminates within $15$
time units, as specified by the following invariance formula:
\[ \G\bigl(T \leq 15 \lvee (\atl{2} \wedge \atm{1})\bigr) \]

\noindent This formula claims that every accessible state in which the
program has not terminated yet (i.e., $\atl{2} \wedge \atm{1}$ does
not hold), can only be observed when the master clock $T$ has not yet
passed $15$. It follows that any state observed later than $T=15$ must
be a termination state.

The proof is presented by the invariance diagram in
Fig.~\ref{fig:exr1}.
\begin{figure}[hbt]
\begin{center}
\leavevmode 
\input exr1.pstex_t
\caption{\label{fig:exr1} Termination of {\sc any-y} within $15$ time units.}
\end{center} 
\end{figure} 

\nin
The assertions are indexed in descending order from $\varp_3$ on top to
$\varp_0$ at bottom in order to keep compatibility with the chain
diagrams introduced in Section~\ref{response}.

Note that no edge in the diagram is labeled by the \tick\ transition.
This implies that all verification conditions involving the \tick\
transition are of the form $\triple(\varp_i,\tick,\varp_i)$ claiming
that the \tick\ transition preserves each of the assertions appearing
in the diagram.

As an example, consider the verification condition claiming that
assertion $\varp_1\colon\ \atl{1}\lwedge T\le 5+ t_1\le 10$ is
preserved by the \tick\ transition.
$$\begin{array}{l}
\underbrace{\atl{1}\lwedge T\le 5+ t_1\le 10}_{\varp_1} \quad \band \\
\multicolumn{1}{r}{\underbrace{\D>0 \lwedge \pi\pri=\pi
\lwedge T\pri = T+\D \lwedge t_1\pri = t_1+\D \lwedge (\atl{1} \to
t_1+\D \le 5) \lwedge \cdots}_{\rotick}} \\
\multicolumn{1}{r}{\implies \quad
\underbrace{
\atlp{1}\lwedge T\pri\le 5+ t_1\pri\le 10}_{\varp_1\pri}}
\end{array}$$
It is not difficult to see that this implication is state-valid.\corn

\end{example}

%\input{example5.tex}
\begin{example}\quad
As a second example, we present a mutual-exclusion
algorithm, due to M. Fischer, which functions properly only due to the
timing constraints associated with the statements. Similar proofs to
the one we will present here are given in \cite{SBM91}, \cite{AL91},
and \cite{MMP}.

The algorithm is presented in Fig.~\ref{fschr} under the name of
program \mtx. By assigning all statements in \mtx\ the uniform time
bounds $[L,U]$, we obtain the timed program \mtxlu.  Assuming that
$2L > U$, we prove that the mutual exclusion property
\[\G\neg(\atl{8} \lwedge \atm{8}) \]
is valid for \mtxlu.
\begin{figure}[hbtp]
\begin{center}
\fbox{
$
\begin{array}{c}
x\colon\ \bold{integer\ where}\ x=0\\
\quad\\
\begin{array}{ccc}
\left[\begin{array}{l}
\ell_0: \ \bold{loop forever do}\\
\quad \left[\begin{array}{ll}
  \ell_1: & \bold{noncritical}\\
  \ell_2: & \bold{skip}\\
  \ell_3: & \bold{while $x\ne 1$ do}\\
  \multicolumn{2}{l}{\quad\left[\begin{array}{ll}
    \ell_4: & \bold{await $x=0$} \\
    \ell_5: & x:= 1	\\
    \ell_6: & \bold{skip}\\
    \ell_7: & \bold{if $x=1$ then} \\
            & \ell_8: \ \bold{critical}
    \end{array}\right]}\\
  \ell_9: & \bold{skip}\\	
  \ell_{10}: & x := 0
  \end{array}\right]
\end{array}\right]
& \quad\PLL\quad &
\left[\begin{array}{l}
m_0: \ \bold{loop forever do}\\
\quad \left[\begin{array}{ll}
  m_1: & \bold{noncritical}\\
  m_2: & \bold{skip}\\
  m_3: & \bold{while $x\ne 2$ do}\\
  \multicolumn{2}{l}{\quad\left[\begin{array}{ll}
    m_4: & \bold{await $x=0$} \\
    m_5: & x:= 2	\\
    m_6: & \bold{skip}\\
    m_7: & \bold{if $x=2$ then} \\
            & m_8: \ \bold{critical}
    \end{array}\right]}\\
  m_9: & \bold{skip}\\	
  m_{10}: & x := 0
  \end{array}\right]
\end{array}\right]\\
\quad \\
-\quad P_1 \quad - & & -\quad P_2 \quad -
\end{array}
\end{array}
$
}
\caption{\label{fschr} Program \mtx, implementing Fischer's protocol.}
\end{center}
\end{figure}

Let us explain informally why mutual exclusion for program \mtx\ is
guaranteed and the role of the \emm{skip} statements at $\ell_6$ and
$m_6$. Assume a violation of mutual exclusion in which process $P_1$
ewntered its critical section (location $\ell_8$) first and, while
$P_1$ is still there, process $P_2$ enters $m_8$. When $P_1$ entered
$\ell_8$, $x$ was $1$. For $P_2$ to enter $m_8$ later, it was
necessary for it to set $x$ to 2 first, which can only be done at
$m_5$. Since after $P_1$ set $x$ to 1 at $\ell_5$, $P_2$ cannot pass
the test for $x=0$ at $m_4$, the only possibility is that $P_2$ kept
waiting at $m_5$ while $P_1$ executed $\ell_6$ and the test at
$\ell_7$. This must have taken $P_1$ at least $2L$, since $L$ is the
lower bound for execution of a statement. However, $P_2$ cannot wait
at $m_5$ for as long as $2L$ because $2L>U$ and no process can delay
the execution of a statement for more than $U$. It follows that the
described scenario in which $P_2$ keep waiting at $m_5$ until $P_1$
enters $\ell_8$ is impossible. The role of the \emm{skip} statement at
$\ell_6$ is therefore to introduce an additional delay of at least $L$
time units.

The \emm{skip} statements at $\ell_9$ and $m_9$ are
necessary to guarantee the response properties of this algorithm, and
we will discuss them in the next section. The \emm{skip} statements at
$\ell_2$ and $m_2$ represent exits out of the noncritical sections (at
$\ell_1$ and $m_1$, respectively) and intention to enter the critical
sections.

The property of mutual exclusion for program \mtx\ can be stated by
the invariance formula
\[\G\neg (\atl{8}\band \atm{8}).\]
A formal proof of this invariant is presented by the invariance
diagram of \Figr{invmtx}.
\begin{figure}[hbt]
\begin{center}
\leavevmode 
\input invmtx.pstex_t
\caption{\label{invmtx} Mutual exclusion for program \mtx.}
\end{center} 
\end{figure}

Note that the edge labeled $\ell_5$ reaches the compound node labeled
$x=1$. By the encapsulation conventions, this is equivalent to having
this edge connect to the basic node labeled $\atl{6..10},\
\atm{0..4,6,7,9,10}$ and the compound node labeled $\atm{5}$. While
reaching this compound node, it should enter some of its contained
basic nodes. The default-entry arrow marks the basic node labeled
$\atl{6},\ t_2\ge t_1$ as the single node entered by the $\ell_5$
edge. Thus, the $\ell_5$ edge connects to the basic node labeled
$\atl{6..10},\ \atm{0..4,6,7,9,10}$ and the one labeled $\atl{6},\
t_2\ge t_1$.
\corn
\end{example}

\subsection*{Completeness of Rule \inv}
Rule \inv\ is complete for verifying invariances of fair transition
systems. This is stated by the following claim.

\begin{claim}
\label{complinv}
If formula $\G p$ is valid over non-zeno \ctss\ $P$, then there exists
an assertion $\varp$ such that premises {\rm I1}-{\rm I3} of rule \inv\
are state-valid.
\end{claim}

\nin{\bf Justification}\quad
The basic idea of the proof is the construction of an assertion
$\accp$ that holds in a state $s$ iff $s$ is $P$-accessible, i.e.,
appears in some run of $P$. We then show that if $\G p$ is $P$-valid,
then the premises of rule \inv\ are state-valid (implying that they
are also $P$-state valid) when taking $\accp$ for $\GF$. Since we are
only proving \emm{relative} completeness, it is enough to show
validity of the premises, assuming an oracle that provides proofs or
otherwise verifies all generally valid assertions.

For the construction of $\accp$, we refer the reader to \cite{Comp91}
or Chapter 2 of \cite{MPB2}. Assume that we have constructed an
assertion $\accp$ such that
\begin{quote}
$s\models \accp$ \qquad iff \qquad $s$ is a $P$-accessible state.
\end{quote}

\nin
We show that $\accp$, when substituted for $\GF$, validates the three
premises of rule \inv.
\begin{itemize}
\item[I1.] $\accp\to p$\\
By our assumption that $\G p$ is $P$-valid, it follows that each state
appearing in some computation satisfies $p$. As $P$ is non-zeno, every
accessible state appears in some computation and, hence, every
accessible state satisfies $p$. Since $\accp$ characterizes precisely
the accessible states, the premise follows.
\item[I2.] $\Theta\to\accp$\\
It is obvious that every state satisfying $\The$ is initial and
is, therefore, $P$-accessible. Consequently, such a state must satisfy
$\accp$, which characterizes all $P$-accessible states.
\item[I3.] $\rot(U,\wtU)\lwedge \accp(U) \quad \to \quad
\accp(\wtU)$,\ for each $\tau\in\cT$ and every values of $U$ and
$\wtU$.\\
Let $U$ and $\wtU$ be two lists of values, which can be viewed as values
of the system variables $V$, such that both $\rot(U,\wtU)$ and $\accp(U)$
are true. We will show that $\accp(\wtU)$ is also true.

Let $s$ and $\wts$ be two arbitrary states such that $s[V]= U$ and
$\wts[V] = \wtU$. Since $\accp(U)=\scT$, we have that $s\models \accp$.
By the meaning of $\accp$, it follows that $s$ is $P$-accessible. From
$\rot(U,\wtU) =\scT$, it follows that $\langle s,\wts \rangle \models
\rot(V ,V\pri)$ and, therefore, that $\wts$ is a $\tau$-successor of
$s$. Since $s$ is accessible, it follows that also $\wts$ is
accessible and, hence $\wts\models \accp$, leading to $\accp(\wtU) = \scT$

\end{itemize}
This concludes the proof of completeness of rule \inv.\corn

\medskip

The following example demonstrates that the assumption that $P$ is
non-zeno is essential to the completeness of the rule.

\begin{example}\quad
Consider the \ctss\ $\Phi_3$ which is presented graphically in
\Figr{phic} and textually in \Figr{phictab}.
\begin{figure}
\begin{center}
\fbox{
\begin{tabular}{ccc}
\quad\\
& \input {phic.pstex_t} & \\
\quad
\end{tabular}
}
\caption{\label{phic} \ctss\ $\Phi_3$.}
\end{center} 
\end{figure}

Clearly, \ctss\ $\Phi_3$ is a possibly-zeno system because, once a run
enters location $\ell_1$, time is bounded and cannot diverge. Thus, a
run entering location $\ell_1$ cannot be extended into a computation
of $\Phi_3$. A result of the fact that no computation of $\Phi_3$ ever
enters location $\ell_1$ is that the invariance formula $\G\atl{0}$ is
$\Phi_3$-valid.
\begin{figure}
$$
\begin{array}{|c|}
\hline
\quad \\[-8pt]
\ \begin{array}{rl}
V\colon & \{\pi\colon \{0,1\}; t,T\colon \mbox{\bf real}\} \\
\Theta\colon & \pi = t = T = 0 \\
\Tau\colon & \{\tau_{01}\}\ \mbox{with transition relations}\\
& \begin{array}{rl}
\tau_{01}\colon & \pi =0 \lwedge \pi\pri = 1 \lwedge t\pri = 0 \lwedge
T\pri = T
\end{array}\\
\Pi\colon & \pi=1 \to t<1
\end{array}\ \\[-8pt]
\quad \\
\hline
\end{array}
$$
\caption{\label{phictab} \ctss\ $\Phi_3$ in textual form.}
\end{figure}
However, examination of the arguments for the soundness of rule \inv\
shows that any assertion $p$, whose invariance is proven by rule \inv,
must hold on all accesible states. The state $\lan \pi\colon 1\ran$ is
accessible for \ctss\ $\Phi_3$ but does not satisfy $\atl{0}$.
Therefore, the $\Phi_3$ validity of $\G\atl{0}$ cannot be proven by
rule \inv.
\corn

\end{example}

%\input{waiting}
\subsection{Verifying Waiting-for Formulas}
The invariance formula $\G q$ states that $q$ holds continuously from
the beginning of the computation to infinity. In comparison, the
waiting-for formula
\[p \quad\entails\quad q\W r\]
also states the continuous holding of $q$ but only for an interval
that is initiated by an occurrence of $p$ and may be terminated by an
occurrence of $r$. Such formulas are useful for expressing timing
properties, where time is measured since an occurrence of an event,
rather than from the beginning of a computation.

\begin{example}\quad
Consider program \cyclic, presented in \Figr{cyclicp}.
\begin{figure}[hbtp]
\begin{center}
\fbox{
$
\begin{array}{c}
x, y\colon\ \bold{integer\ where}\ x=y=0\\
\quad\\
\begin{array}{ccc}
%P_1::\ \ 
\left[\begin{array}{l}
\ell_0: \ \bold{loop forever do}\\
\quad\left[\begin{array}{ll}
\ell_1: & \bold{while}\ x=0\ \bold{do}\\
\multicolumn{2}{l}{
\quad\ell_2:\ y:=y+1}\\
\ell_3: & x := 1
\end{array}\right] 
\end{array}\right]
& \quad\PLL\quad &
%P_2::\ \ 
\left[\begin{array}{l}
m_0: \ \bold{loop forever do}\\
\quad\left[\begin{array}{ll}
m_1: & x:= 1\\
m_2: & \bold{await $x=0$}
\end{array} \right]
\end{array} \right]\\
\quad\\
-\quad P_1 \quad - & & -\quad P_2 \quad -
\end{array}
\end{array}
$
}
\caption{\label{cyclicp} Program \cyclic.}
%
\end{center}
\end{figure}
This program can be viewed as a generalization of program \anyy, in
which the basic interaction between processes $P_1$ and $P_2$ is
embedded within an endless loop.

Since program \cyclic\ never terminates, a relevant time-related
property is that from any state in which we observe $P_1$ at $\ell_1$
and $P_2$ at $m_1$, $P_1$ will reach location $\ell_3$ within at most
$15$ time units.

This property can be stated by the following waiting-for formula:
\[\atl{1}\lwedge \atm{1}\lwedge x=0 \lwedge T=a
\quad\entails\quad (T\le a+15)\W\atl{3}.\]
This formulas uses a rigid variable $a$ to record the time at the
initial observation state. It claims that from the time we observe a
state satisfying $\atl{1}\lwedge \atm{1}\lwedge x=0$, time will not
progress by more than 15 units before we observe $P_1$ at $\ell_3$.

To facilitate the expression of such properties, we introduce the
abbreviation
\[\Ta\colon\quad T-a.\]
With this abbreviation we can write the property of ``reaching
$\ell_3$ within $15$ time units'' as
\[\atl{1}\lwedge \atm{1}\lwedge x=0 \lwedge \Ta=0
\quad\entails\quad (\Ta\le 15)\W\atl{3}.\]
The $\Ta$ abbreviation allows us to view $\Ta$ as a special timer,
reset at the initial point of observation.
\corn
\end{example}

\ssb{A Waiting-for Rule}
To establish the $P$-validity of a waiting-for formula, we may use
rule \wait, presented in \Figr{wait}.
\begin{figure}
\LLbox{\boxline{\qquad For assertions $p$, $q$, $\varp$, and $r$,}
\Ruleb{U1.\quad $\varp\ \rar\ q$\cr
\noalign{\vskip 5pt}
U2.\quad $p\ \rar\ r\lvee \varp$\cr
\noalign{\vskip 5pt}
U3.\quad $\rho_\tau\band\varp\ \ \rar\ \ r\pri \lvee \varp\pri$\quad
for every $\tau\in\Taut$\cr
\Bar
$\qquad\hskip 1pt\Sys\models p\quad\entails\quad q\W r$\cr}\lskip}
\ctrlined{wait}{Rule \wait\ (waiting-for) applied to \ctss\ $\Sys$.}

\ssb{Waiting Diagrams}
Proofs by rule \wait\ can be succinctly represented by special
verification diagrams to which we refer as \emm{waiting diagrams}.

A waiting diagram is a directed labeled graph constructed as follows:
\begin{itemize}
\item \emm{Nodes} in the graph are labeled by assertions $\varphi_0,
\ldots, \varphi_m$. We will often refer to a node by the assertion
labeling it.

The node labeled by $\varp_0$ is called the \emm{terminal node} and is
graphically identified by being drawn with bold-face boundaries.

\item \emm{Edges} in the graph represent transitions between
assertions. Each edge is labeled by the name of a transition in the
program. We refer to an edge labeled by $\tau$ as a \emm{$\tau$-edge}.
No edges depart form the terminal node.

\item Some of the nodes are designated as \emm{initial nodes}. They
are annotated by an entry arrow.
\end{itemize}

\ssb{Verification Conditions Implied by a Chain Diagram}
Consider a nonterminal node labeled by assertion
$\varp$. Let $\tau\in \Taut$ be a transition and let
$\varp_1,\ldots,\varp_k$, $k\ge 0$, be the successors of $\varp$ by
edges labeled with $\tau$ (possibly including $\varp$ itself).
With each such node and transition, we associate the
following verification condition:
\[\rho_\tau\band\varp\ \ \rar\ \ \varp\pri
\bor \varp\pri_1\ \bor\ \cdots\ \bor\ \varp\pri_k.\]
In particular, if $k=0$ (i.e., $\varp$ has no $\tau$-successors), the
associated verification condition is
\[\rho_\tau\band\varp\ \ \rar\ \ \varp\pri.\]

\ssb{Valid Waiting Diagrams}
A waiting diagram is said to be \emm{valid over \ctss\ $P$}
($P$-\emm{valid} for short) if all the verification conditions
associated with the diagram are $P$-state valid.

The consequences of having a valid chain diagram are stated in the
following claim:

\begin{claim}
\label{validwaitd}
If $D$ is a $P$-valid waiting diagram with nodes
$\varp_0,\ldots,\varp_m$, then
\[P\models {\bigvee_{j=0}^m}\ \varp_j\ \ \Rar\ \ (\bigvee_{j=0}^m\W\varp_0\]
If, in addition, $\varp_0= r$,
\[\mbox{{\rm U1:}}
\quad \displaystyle{\bigvee^m_{j=0}}\ \varp_j\ \ \rar\ \ q,
\qquad\mbox{and}\qquad
\mbox{{\rm U2:}}
\quad p\ \ \rar\ \ \displaystyle{\bigvee^m_{j=0}}\ \varp_j,
\]
then we can conclude:
\[P\models p\ \ \Rar\ \ q\W r.\]
\end{claim}

\nin{\bf Justification}\quad
We observe first that the verification conditions associated with a
waiting diagram imply premise U3 of rule \wait, when we take $\varp$
to be $\varp_1\lvee\cdots\lvee\varp_m$ and $r$ to be $\varp_0$.

If we further take $p= \varp_0\lvee\cdots\lvee\varp_m$ and
$q= \varp_1\lvee\cdots\lvee\varp_m$, we find that
premises U1 and U2 hold trivially. This yields the first part of the
claim.

The second part of the claim considers different $p$ and $q$ but
explicitly requires that premises U1 and U2 are $P$-valid. The
conclusion follows by rule \wait.\corn

\begin{example}\quad
In \Figr{wait1}, we present a waiting diagram that establishes the
property
\[\atl{1}\lwedge \atm{1}\lwedge x=0 \lwedge \Ta=0
\quad\entails\quad (\Ta\le 15)\W\atl{3}\]
for program \cyclic.
\begin{figure}[hbt]
\begin{center}
\leavevmode 
\input wait1.pstex_t
\caption{\label{wait1} A waiting diagram, establishing the formula}
$\atl{1}\lwedge \atm{1}\lwedge x=0 \lwedge \Ta=0 \quad\entails\quad
(\Ta\le 15)\W\atl{3}$
\end{center} 
\end{figure}
\corn
\end{example}
